Author’s response to reviews

Title: An evaluation of methods for the isolation of nontuberculous mycobacteria from patients with cystic fibrosis, bronchiectasis and patients assessed for lung transplantation

Authors:

Dominic Stephenson (Dominic.Stephenson@nuth.nhs.uk)
Audrey Perry (Audrey.Perry@nuth.nhs.uk)
Martin Appleby (Martin.Appleby@nuth.nhs.uk)
David Lee (dave.lee@nuth.nhs.uk)
John Davison (John.Davison2@nuth.nhs.uk)
Annette Johnston (annette.johnston@northumbria.ac.uk)
Amanda Jones (amanda.l.jones@northumbria.ac.uk)
Andrew Nelson (andrew3.nelson@northumbria.ac.uk)
Stephen Bourke (Stephen.Bourke@nuth.nhs.uk)
Matthew Thomas (Matthew.Thomas3@nuth.nhs.uk)
Anthony De Soyza (anthony.de-soyza@newcastle.ac.uk)
James Lordan (Jim.Lordan@nuth.nhs.uk)
Joanna Lumb (Joanna.Lumb@nuth.nhs.uk)
Ali Robb (Ali.Robb@nuth.nhs.uk)
Julie Samuel (Julie.Samuel@nuth.nhs.uk)
Katherine Walton (Kathy.Walton@nuth.nhs.uk)
John David Perry (john.perry@nuth.nhs.uk; homerresearch09@gmail.com)

Version: 4 Date: 31 Dec 2018

Author’s response to reviews:

Editor Comments:
1. In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

   Author response: Done.

2. Please move the List of Abbreviations and the Declarations to after the Conclusions section, before the References.

   Author response: Done.

3. Please amend the following headings in your manuscript, as indicated:
   - “Introduction”: please change this to “Background”

   - “Materials and Methods”: please change this to “Methods”

   Author response: Done.

4. Please remove the cover letter from the file inventory, as it is no longer needed at this stage.

   Author response: Done.

5. Please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

   Author response: Done